July 13, 2021

Re: JoVE62720R1

"A Novel Intact Pericardium Ischemic Rodent Model"

Response to Reviewers

Dear Editor-in-Chief,

Thank you for your email regarding our submission to *JoVE*. We appreciate the comments and feedback provided by the Editorial Board. We have revised our manuscript to address the issues highlighted by the reviewers and editors. The following is an itemized response:

Editorial comments:

Changes to be made by the Author(s):

1. Please thoroughly proofread the manuscript.

Response: Thank you.

2. The manuscript is formatted to match the journal style. Please review.

Response: Thank you.

- 3. Please revise the following lines to avoid previously published work: 46-48, 54-55, 64-65, 73-78 Response: We have reviewed these lines in the copy provided and we do not make reference to previous published work. Please advise how to proceed.
- 4. Please note that your protocol will be used to generate the script for the video and must contain everything that you would like shown in the video. Please ensure you answer the "how" question, i.e., how is the step performed? Alternatively, add references to published material specifying how to perform the protocol action. There should be enough detail in each step to supplement the actions seen in the video so that viewers can easily replicate the protocol.

Response: Thank you. We believe this protocol is now ready to be filmed. There is sufficient detail enclosed that can facilitate the adoption and reproduction of this protocol.

5. Please format the References as per JoVE style. Ref 1 and 2 has been done. Please follow the same formatting for all.

Response: Thank you, we have updated the formatting of the references.

Reference format

[Lastname, F.I., LastName, F.I., LastName, F.I. Article Title. Source. Volume (Issue), FirstPage – LastPage (YEAR).] For more than 6 authors, list only the first author then et al.

6. Please mark the specific changes made by you in the manuscript in response to all the comments.

Response: Thank you, all changes are in tracks.

7. Additional comments are in the attached manuscript.

Response: Thank you, we have addressed all the comments.

We hope to have been able to offer satisfactory modifications and explanations to the issues highlighted by the Editorial Board and would like to hereby submit the manuscript for consideration on a de-novo revision basis.

Please find attached a copy of the revised manuscript and a copy containing the marked changes.

Thank you for your time and consideration.

Sincerely yours,

Justin Deniset, PhD

Assistant Professor
Libin Cardiovascular Institute
Dept. of Physiology and Pharmacology and Cardiac Sciences
Cumming School of Medicine
University of Calgary
3330 Hospital Drive NW
Calgary, AB, T2N 4N1, Canada
Ph: (204) 995-5110